## Threat to low-risk birth environments

Emma Smith discusses the effect of the current staffing crisis on low-risk birth environments, and explores how the challenges that they face may be overcome

inwardly sigh as I receive the call at the end of my shift that the birth centre where I work is having to be diverted yet again. While the night team head to the labour ward because of high acuity. I feel sad and deflated as I turn off lights, divert the phones and place a 'closed' sign on the birth centre door. On my drive home, I start to imagine women going into labour overnight and heading into the unit, keenly anticipating the birth centre as the place they will labour. I imagine the lump in the throat swallowed back, the shoulders drooping and the disappointed eyes as they are informed of its temporary closure for the shift and their questions, 'is there no chance of it being open at all?' or 'is there a pool available?' Frustration takes hold of me, and a lump threatens to form in my throat when I imagine the answers they are likely to receive: 'there aren't enough midwives to staff the birth centre tonight' and 'I'm afraid the pool room is already in use'. I think back to the conversations I have had with women who tell me the feelings of dread and panic when they heard they could not use a pool, the birth centre was not open, or there was no community midwife available to attend their planned home birth. These familiar cases have led me to a conclusion: low-risk labour care is at threat with detrimental consequences.

### **Current staffing crisis**

While all areas of maternity services are currently struggling, the past 3 years in the

### **Emma Smith**

Registered midwife, Abbey Birth Centre, Ashford and St Peters Hospital NHS Foundation Trust emma.smith179@nhs.net

UK have placed the provision of low-risk labour care at a serious disadvantage (National Institute for Health and Care Excellent (NICE), 2023). Many units found that they were unable to safely provide a home birth service during the COVID-19 pandemic and made the difficult decision to suspend it. The maternity staffing crisis during that time, and following it, has meant that many units are having to divert their birth centre midwives to where acuity is at its highest and their birth centres are temporarily closed (NICE, 2023). Women are finding themselves limited for choice and are forced to give birth in a setting not of their choosing. Other than being extremely disappointing for these women, there are numerous disadvantages to labouring in high-risk environments if not clinically indicated.

#### **Birthplace study findings**

The findings of the Birthplace study (Birthplace in England Collaborative Group, 2011) confirmed the safety and numerous benefits for low-risk women who labour in a midwifery-led setting, such as home births and midwifery-led units. It was noted that for nulliparous and multiparous women who planned birth in midwifery-led units, the rate of interventions such as caesarean sections and episiotomies were reduced, and they had a higher rate of spontaneous vaginal births (Birthplace in England Collaborative Group, 2011). As much as interventions are important and sometimes necessary, many interventions increase the risk of further complications for women and babies.

### Interventions and their risks

Women are more likely to receive continuous fetal monitoring or

cardiotocography on an obstetric-led unit, which is associated with a rise in interventions, appearing to make little difference in preventing adverse neonatal outcomes (Goer and Romano, 2012; Alfirevic et al, 2017; Politi et al, 2023). As a result of its association with increased interventions, NICE (2022) does not recommend routine cardiotocography for low-risk women in labour. However, cardiotocography is still used unnecessarily (Jepsen et al, 2022).

Women are more likely to have higher use of water immersion in midwifery-led units and in home settings (Baczek et al, 2020; Burns et al, 2022), but those who labour in obstetric-led units are less likely to access a birthing pool (Milosevic et al, 2019). Numerous benefits to both mother and babies are evident when water immersion is used in labour, such as reduced epidural and opioid use, reduced episiotomies, higher rates of spontaneous vaginal births, a shorter labour and increased maternal satisfaction (Henderson et al, 2014; Lukasse et al, 2014; Ulfsdottir et al, 2018; Burns et al, 2022; Reviriego-Rodrigo et al, 2023). If being on an obstetric-led unit equals reduced access to water immersion, then it is safe to say that more women will opt for epidurals. I have heard women speak of their dismay at hearing that they cannot access a birthing pool on an obstetricled unit and having fears that they will struggle to cope without it, meaning that they opt for an epidural.

While epidurals can offer a fantastic form of analgesia and do have benefits, they carry disadvantages. Epidurals increase the rate of instrumental birth (Antonakou and Papoutsis, 2016; Anim-Somuah et al, 2018; Penuela et al, 2019), in turn increasing the incidence of perineal trauma, postpartum haemorrhage

2024 MA Healthcare Ltd

Caesarean sections in themselves carry risks such as postpartum haemorrhage, infection, damage to surrounding internal organs, and deep-vein thrombosis (NHS, 2023). Having an emergency caesarean section increases the likelihood of having one in subsequent pregnancies, increasing the risk of uterine rupture, scar tissue and internal adhesions, erratic placentation, postpartum haemorrhage and stillbirth (O'Neill et al, 2014; Royal College of Obstetricians and Gynaecologists, 2015b). Emerging evidence indicates that caesarean sections may also affect the baby's short and long-term health (Moncrieff, 2018), potentially causing respiratory problems, altered immunity and obesity (Sandall et al, 2018). With current rates of caesarean sections soaring (World Health Organization, 2021), the unavailability of low-risk birthing environments is surely only contributing to this number.

### Effects of intervention on maternal mental health

Instrumental births and emergency caesarean sections are also associated with poor maternal mental health, particularly with conditions such as childbirth-related post-traumatic stress disorder (Kitzinger, 2006; Dekel et al, 2019), which can have a detrimental effect on a mother's relationship with their baby and other family relationships (Kitzinger, 2006; Skinner and Dietz, 2014). Poor bonding between mother and child can affect a baby's overall wellbeing and future health (Cook et al, 2018; Royal College of Midwives, 2019a). Considering that interventions such as operative births are associated with mental health problems (Dekel et al, 2019) and that the Mothers

and Babies: Reducing Risk Through Audits and Confidential Enquires Across The UK(2022) report listed suicide as one of the leading causes of maternal death, more priority should be placed on protecting low-risk birth.

# Satisfaction rates in midwifery-led units and home births

Midwifery-led units and home environments are associated with higher rates of maternal satisfaction and an overall positive birthing experience, in contrast to obstetric-led units (McNeils, 2013; Cross-Sudworth et al, 2018; Bączek et al, 2020). The power and importance of oxytocin in labour is known, as well as how factors such as bright lights and a 'clinical looking' environment can

Staff have found the closure of midwifery-led units and suspension of home births frustrating and disheartening as women are unable to access the care they deserve

'frighten' oxytocin away (Goldkuhl et al, 2022). These feelings of satisfaction and relaxation may contribute to a surge in and maintenance of natural oxytocin, hopefully reducing the need for synthetic oxytocin and its associated interventions.

It seems that the 'cascade of intervention,' a phrase that many midwives will be familiar with, may be triggered by women being unable to access midwifery-led units and home births. The closure of low-risk birthing environments clearly has an impact on the health of women and babies. These impacts have a clear ripple effect, potentially affecting a mother and baby's health far into the future.

### **Effects on midwives**

There are potential effects to mothers and babies, but does limiting low-risk

birthing environments affect midwives? The WHELM study (Royal College of Midwives, 2019b) found that a lack of staffing and resources contributed to poor mental health among midwives. McGrory et al (2022) highlighted how the suspension of home births and midwifery-led units closure contributed to staff stress.

Anecdotally, I know that staff have found the closure of midwifery-led units and suspension of home births frustrating and disheartening as women are unable to access the care they deserve. While low-risk labour care may not be every midwife's passion or area of interest, not being able to provide care they feel is important would no doubt cause frustration in any midwife. As a midwife on a midwifery-led unit, I have found that my feelings of burnout start to surface when women are not able to access the birth centre. I can become despondent and workplace satisfaction takes a dip. Mental fatigue can set in and the wearying feeling of 'oh no, not again' surges when the midwifery-led unit is frequently on divert. It would be interesting to know if there is a link between rising levels of intervention in maternity care and poor mental health among midwives.

### **Effects on student midwives**

Restricting low-risk birthing environments also places student midwives at a clear disadvantage (Warwick, 2012; Nash, 2020), as they are less likely to have exposure to 'normal' labour and birth without interventions. I often describe the care provided on a midwifery-led unit or at a home birth as 'the bread and butter of midwifery' to students who feel nervous about working in these settings. This is the care that has been around for centuries; that trust in a woman's body, that reliance on your own senses of listening, touching, watching and trying tricks as old as the hills to enhance the physiological process. While that is not to say that midwives do not or cannot use these skills in a high-risk environment, much more reliance may be placed on machines and interventions. What better way for a student to learn these basics first and then have a solid

Many students have expressed to me poor exposure to and lack of confidence over intermittent auscultation and feel more confident with continuous fetal monitoring. Several students I have come across are close to qualifying and have never witnessed a water birth. This is concerning, considering that we are trying to develop midwives who should feel confident and be able to deliver safe care in all settings.

Are we unwittingly raising a generation of midwives not confident with providing low-risk labour care? Throughout history, midwives have been experts in physiological labour and birth. What does this mean for the future of maternity services and the wider health of women and babies? Warwick (2012) stated that if practitioners lack confidence in low-risk labour care, then it affects the likelihood of women being offered it. Will women continue to be provided with the choice they legally have a right to?

### What can be done?

On taking a step back and surveying the sobering evidence, it is easy to become downhearted when considering where maternity services are currently at. More midwives and more funding come to mind as the immediate solution. While this is vital, changes cannot happen overnight. So what can be done in the meantime? Is there a way we can still protect low-risk birth and its environment?

Sadly, because of staffing problems across the country, there are times that despite everyone's best efforts, a midwifery-led unit may be closed or home birth services not always available. There is a phrase used by the army, 'adapt and overcome', and this could not be truer for the midwifery profession now. Midwives may find themselves called to be more adaptable and create a low-risk environment in a high-risk one, calling for a heavy dose of creativity. To do this, the following questions may need to be asked:

- Does the bed need to be in the centre of the room?
- Do the lights need to be so bright, or are there softer ones available?

 If accessing the pool is not an option, can the labouring woman still access some form of hydrotherapy, such as a shower or bath?

We should not forget that 'midwifery is an art'. Making changes may not come easily, so along with creativity, I am prescribing an even stronger dose of courage; something we are encouraged to uphold in the NHS Professionals (2023) '6 Cs of care'. It takes courage to make changes, to take a woman off an unnecessary cardiotocography, to change an obstetric-led unit room, to challenge an obstetrician, and to speak up when a woman is not being listened to or has not been provided with choice. Support from the whole multidisciplinary team is required to make these changes

Midwives may find themselves called to be more adaptable and create a low-risk environment in a high-risk one, calling for a heavy dose of creativity

happen. It takes senior midwives instilling confidence in other midwives to facilitate physiological birth. It takes obstetricians recognising the 'sacredness' of the birthing room, respecting the importance of an undisturbed birthing environment, and minimising intervention where possible.

### **Conclusions**

The fight is on for low-risk birth, maybe in a way it never has been before. There have always been challenges to facilitating low-risk birth, but the threat seems to have increased. The unavailability of low-risk birthing environments has more of an impact on maternal and infant health than we realise, with consequences that may stretch far into the future. It also threatens the wellbeing of midwives, potentially creating midwives unskilled in low-risk labour. It might sound like stating

the obvious to say that urgent change is needed. It is realistic to say that these changes are unlikely to happen quickly. However, we need to take courage in our capability as midwives to facilitate birth, no matter the setting, and our ability to create, challenge, and protect. Now is the time to start raising our voices and shouting about this important aspect of maternity care for the sake of women and babies. BJM

Alfirevic Z, Devane D, Gyte GMI, Cuthbert A. Continuous cardiotocography (CTG) as a form of electronic fetal monitoring (EFM) for fetal assessment during labour. Cochrane Database Syst Rev. 2017;2(2):CD006066. https://doi.org/10.1002/14651858.cd0060666.pub3

Anim-Somuah M, Smyth RM, Cyna AM, Cuthbert A. Epidural versus non-epidural or no analgesia for pain management in labour. Cochrane. 2018;5(5):CD000331. https://doi. org/10.1002/14651858.cd000331.pub4

Antonakou A, Papoutsis D. The effect of epidural analgesia on the delivery outcome of induced labour: a retrospective case series. Obstet Gynecol Int. 2016;2016:5740534. https://doi.org/10.1155%2F2016%2F5740534

Bączek G, Tataj-Puzyna U, Sys D, Baranowska B. Freestanding midwife-led units: a narrative review. Iran J Nurs Midwifery Res. 2020;25(3):181–188. https://doi.org/10.4103/ ijnmr.IJNMR\_209\_19

Birthplace in England Collaborative Group.

Perinatal and maternal outcomes by planned place of birth for healthy women with low risk pregnancies: the Birthplace in England national prospective cohort study. BMJ. 2011;343:1–13. https://doi.org/10.1136/bmj.d7400

Burns E, Feeley C, Hall PJ, Vanderlaan J. Systematic review and meta-analysis to examine intrapartum interventions, and maternal and neonatal outcomes following immersion in water during labour and waterbirth. BMJ Open. 2022;12(7):e056517. https://doi.org/10.1136/bmjopen-2021-056517

Cook N, Ayers S, Horsch A. Maternal posttraumatic stress disorder during the perinatal period and child outcomes: a systematic review. J Affect Disord. 2018;225:18–31. https://doi.org/10.1016/j.jad.2017.07.045

Cross-Sudworth F, Hindley J, Cheatham C, Clarke P, McAree T. Creating a dedicated homebirth service: results of a 3-year pilot. Br J Midwifery. 2018;26(3):164–170. https://doi.org/10.12968/bjom.2018.26.3.164

Dekel S, Ein-Dor T, Berman Z, Barsoumian IS, Agarwal S, Pitman RK. Delivery mode is associated with maternal mental health following childbirth. Arch Women Ment

Health. 2019;22(6):817-824. https://doi.

Goer H. Do epidurals increase cesareans? 2022. https://hencigoer.com/do-epidurals-increasecesareans/ (accessed 26 October 2023)

Goer H, Romano A. Optimal care in childbirth. London: Pinter & Martin Ltd; 2012

Goldkuhl L, Dellenborg L, Berg M, Wijk H, Nilsson C. The influence and meaning of the birth environment for nulliparous women at a hospital-based labour ward in Sweden: an ethnographic study. Women Birth. 2022;35(4):337-347. https://doi.org/10.1016/j. wombi.2021.07.005

Gurol-Urganci I, Cromwell DA, Edozien LC et al. Third- and fourth-degree perineal tears among primiparous women in England between 2000 and 2012: time trends and risk factors. BJOG. 2013;120(12):1516-1525. https://doi. org/10.1111/1471-0528.12363

Henderson J, Burns EE, Regalia AL, Casarico G, Boulton MG, Smith LA. Labouring women who used a birthing pool in obstetric units in Italy: prospective observational study. BMC Pregnancy Childbirth. 2014;14(14):17. https:// doi.org/10.1186/1471-2393-14-17

Jepsen I, Blix E, Cooke H, Adrian SW, Maude R. The overuse of intrapartum cardiotocography (CTG) for low-risk women: an actor-network theory analysis of data from focus groups. Women Birth. 2022;35(6):593-601. https://doi. org/10.1016/j.wombi.2022.01.003

Kitzinger S. Birth crisis. Abingdon: Routledge; 2006

ijerph192013000

Lukasse M, Rowe R, Townend J, Knight M, Hollowell J. Immersion in water for pain relief and the risk of intrapartum transfer among low risk nulliparous women: secondary analysis of the Birthplace national prospective cohort study. BMC Pregnancy Childbirth. 2014;14(1):60.

https://doi.org/10.1186/1471-2393-14-60

McGrory S, Neill RD, Gillen P et al. Self-reported experiences of midwives working in the UK across three phases during COVID-19: a crosssectional study. Int J Environ Res Public Health. 2022;19(20):13000. https://doi.org/10.3390/

McNeils M. Women's experiences of care during labour in a midwifery-led unit in the Republic of Ireland. Br J Midwifery. 2013;21(9):622-631. https://doi.org/10.12968/bjom.2013.21.9.622

Milosevic S, Channon S, Hunter B et al. Factors influencing the use of birth pools in the United Kingdom: perspectives of women, midwives,

and medical staff. Midwifery. 2019;79:102554. https://doi.org/10.1016/j.midw.2019.102554

Moncrieff G. Can continuity bring birth back to women and normality back to midwives? Br J Midwifery. 2018;26(10):642-650. https://doi. org/10.12968/bjom.2018.26.10.642

Mothers and Babies: Reducing Risk Through Audits and Confidential Enquires Across The UK. Missing voices: saving lives, improving mothers' care: lay summary 2022. 2022. https:// www.npeu.ox.ac.uk/assets/downloads/ mbrrace-uk/reports/maternal-report-2022/ MBRRACE-UK\_Maternal\_Report\_2022\_-\_ Lay\_Summary\_v10.pdf (accessed 14 May 2024)

Nash K. Is it time for a more holistic assessment of labour progress? Br J Midwifery. 2020;28(8):457-459. https://doi.org/10.12968/ bjom.2020.28.8.457

National Institute For Health and Care Excellence. Fetal monitoring in labour. 2022. https://www. nice.org.uk/guidance/ng229 (accessed 14 May 2024)

National Institute for Health and Care Excellence. Intrapartum care: section 1.3 planning place of birth, 1.3.7 low risk multiparous women. 2023. https://www. nice.org.uk/guidance/ng235/chapter/ Recommendations#:~:text=Advise%20 low%2Drisk%20multiparous%20 women,compared%20with%20an%20 obstetric%20unit. (accessed: 14 May 2024)

NHS. Caesarean section: risks. 2023. https://www. nhs.uk/conditions/caesarean-section/risks/ (accessed 6 October 2023)

NHS Professionals. The 6 Cs of care. 2023. https:// www.nhsprofessionals.nhs.uk/nhs-staffing-poolhub/working-in-healthcare/the-6-cs-of-care (accessed 14 May 2024)

O'Neill SM, Agerbo E, Kenny CL et al. Cesarean section and rate of subsequent stillbirth, miscarriage, and ectopic pregnancy: a Danish register- based cohort study. PLoS Med. 2014;11(7):e1001670. https://doi. org/10.1371%2Fjournal.pmed.1001670

Penuela I, Isasi-Nebreda P, Almeida H, López M, Gomez-Sanchez E, Tamayo E. Epidural analgesia and its implications in the maternal health in a low parity comunity. BMC Pregnancy Childbirth. 2019;19(1):52. https:// doi.org/10.1186/s12884-019-2191-0

Peters LL, Thornton C, de Jonge A et al. The effect of medical and operative birth interventions on child health outcomes in the first 28 days and up to 5 years of age: a linked data populationbased cohort study. Birth. 2018;45(4):347-357. https://doi.org/10.1111/birt.12348

Politi S, Mastroroberto L, Ghi T. The time has come for a paradigm shift in obstetrics' medicolegal litigations. Eur J Obstet Gynecol Reprod Biol. 2023;284:1-4. https://doi.org/10.1016/j. ejogrb.2023.02.018

Reviriego-Rodrigo E, Ibargoyen-Roteta N, Carreguí-Vilar S et al. Experiences of water immersion during childbirth: a qualitative thematic synthesis. BMC Pregnancy Childbirth. 2023;23(1):395. https://doi.org/10.1186/ s12884-023-05690-7

Royal College of Midwives. Parental emotional wellbeing and infant development. 2019a. https://www.rcm.org.uk/media/4645/ parental-emotional-wellbeing-guide.pdf (accessed 14 May 2024)

Royal College of Midwives. Work, health, and emotional lives of midwives in the United Kingdom. 2019b. https://www.rcm.org.uk/ media/2924/work-health-and-emotional-livesof-midwives-in-the-united-kingdom-the-ukwhelm-study.pdf (accessed 14 May 2024)

Royal College of Obstetricians and Gynaecologists. The management of third-and fourth-degree perineal tears: green-top guideline no. 29. 2015a. https://www.rcog.org.uk/media/5jeb5hzu/gtg-29.pdf (accessed 14 May 2024)

Royal College of Obstetricians and Gynaecologists. Birth after previous caesarean birth greentop guideline no. 45. 2015b. https://www. rcog.org.uk/media/kpkjwd5h/gtg\_45. pdf#:~:text=The%20risk%20of%20 perinatal%20death%20with%20ERCS%20 is,reduced%20with%20a%20preoperative%20 course%20of%20antenatal%20corticosteroids. (accessed 14 May 2024)

Royal College of Obstetricians and Gynaecologists. Assisted vaginal birth: green-top guideline no. 26. 2020. https://www.rcog.org.uk/guidance/ browse-all-guidance/green-top-guidelines/ assisted-vaginal-birth-green-top-guidelineno-26/ (accessed 14 May 2024)

Sandall J, Tribe RM, Avery L et al. Short-term and long-term effects of caesarean section on the health of women and children. Lancet. 2018;392(10155):1349-1357. https://doi. org/10.1016/S0140-6736(18)31930-5

Skinner EM, Dietz HP. Psychological and somatic sequelae of traumatic vaginal delivery: a literature review. Aust N Z J Obstet Gynaecol. 2015;55(4):309-314. https://doi.org/10.1111/ aio 12286

Ulfsdottir H, Saltvedt S, Georgsson S. Waterbirth in Sweden – a comparative study. Acta Obstet Gynecol Scand. 2018;97(3):341-348. https:// doi.org/10.1111/aogs.13286

Warwick C. Outcomes by planned place of birth: implications of the Birthplace Study. Br J Midwifery. 2012;20(1):20-21. https://doi. org/10.12968/bjom.2012.20.1.20

World Health Organization. Caesarean section rates continue to rise, amid growing inequalities in access. 2021. https://www.who.int/news/ item/16-06-2021-caesarean-section-ratescontinue-to-rise-amid-growing-inequalities-inaccess (accessed 14 May 2024)